RESEARCH



A nonsmooth dynamics framework for simulating frictionless spatial joints with clearances

Ekansh Chaturvedi^{1,2} · Corina Sandu¹ · Adrian Sandu²

Received: 21 September 2023 / Accepted: 1 February 2024 © The Author(s) 2024

Abstract

Real-world multibody systems do not have ideal joints; most joints have some clearance. The clearance allows the connected bodies to undergo a misalignment, and the resulting dynamics is governed by the contacts thus formed. Two approaches are typically taken to deal with contacts: the commonly used continuous dynamics approaches assume the Hertzian nature of the contact modeled by nonlinear unilateral spring-damper elements; while the nonsmooth dynamics approach results in a complementarity problem. This paper employs a nonsmooth dynamics approach to develop a coherent framework for the simulation of multibody systems having frictionless joints with clearances. Because clearances are of small magnitude relative to the dimensions of the mechanical components, the nature of the contact in the joints is assumed to be inelastic. Using this assumption and the general nonsmooth dynamics framework, the parametric formulations for cylindrical, prismatic, and revolute joints with clearances are derived. The equations of motion are formulated, and their time-discretized counterparts are cast as a nonlinear programming problem. The proposed scheme also enforces normalization constraint on Euler parameters in contrast to state-ofthe-art methods that is conducive to stability of the solution for a suitable range of step sizes. In addition, a variable time-stepping scheme that includes the step size as an extra variable in the optimization is introduced and its stability properties are discussed. The versatility of the proposed framework is demonstrated through numerical experiments.

Keywords Nonsmooth dynamics \cdot Nonlinear dynamics \cdot Joints with clearances \cdot Contact dynamics \cdot Constraints

C. Sandu csandu@vt.edu

E. Chaturvedi ekanshchat96@vt.edu

A. Sandu sandu@cs.vt.edu

¹ Terramechanics, Multibody and Vehicle Systems Laboratory, Department of Mechanical Engineering, Virginia Tech, Blacksburg, 24060, VA, USA

² Computational Science Laboratory, Department of Computer Science, Virginia Tech, Blacksburg, 24060, VA, USA

1 Introduction

Modern methods for simulating multibody dynamics include algebraic constraint equations to represent mechanical joints. Such equations enforce absolute alignment of the connected bodies in the desired direction of relative motion, and thus represent "ideal joints". However, real mechanical systems have joints with clearances, and the condition of absolute alignment is not always followed. In real-world applications, the motion is governed by intermittent formation of one or multiple contacts. The resulting contact forces are highly nonlinear phenomena that are challenging to model.

Multibody dynamics studies mechanical systems consisting of several bodies that are interconnected with kinematic constraints. These constraints, in the form of mathematical equations, restrict their relative movement and are subjected to the acting external forces that may include gravitational forces, time dependent forces, or contact forces generated by the contacts between the bodies. Indeed, these kinematic constraints enforce absolute alignment of the interacting bodies in a specific direction of the relative motion. However, in the real-world systems, the mechanical components also have clearances and gaps, and the condition of absolute alignment cannot be always expressed by kinematic constraint equations. In such cases, the motion is governed by the formation of contacts at one or multiple points. Consequently, a contact-evaluation procedure must be able to model the dynamics of the contact between compliant surfaces.

The foundations of pure elastic contact models were laid by Heinrich Rudolf Hertz [1]. He concluded that a contact area can be roughly assumed to be elliptical, and no energy dissipation was considered for the contact to be elastic. This led several researchers to develop more advanced contact-force models that take this energy dissipation into account. Hertz's contact law relates the contact force to a nonlinear power function of the indentation and can be expressed as

$$F_c = K \,\delta^n,\tag{1}$$

where *K* is the contact-stiffness coefficient and δ is the relative indentation between the bodies. The lack of energy dissipation can be overcome by including hysteresis damping after the evaluation of contact force. The model developed by Kelvin and Voigt [2] includes energy dissipation, where contact is modeled as a linear spring and a linear damper element as follows:

$$F_c = K\delta + D\dot{\delta},\tag{2}$$

where *D* represents the damping coefficient and δ is the relative normal velocity of indentation. However, the Kelvin and Voigt model does not address the nonlinearity of the contact formation process. Hence, a popular general method for evaluating contact forces has the form

$$F_c = K\delta^n + \chi\delta^n\delta. \tag{3}$$

Several contact force models were formulated based on different methodologies to evaluate the damping coefficient χ and the deformation δ . Some of the most popular contact models include [3–9]. Further, recent rich literature reviews [10–12] provide a comprehensive review of major continuous contact models worked out hitherto.

Besides contact force evaluation, contact detection is another challenge. Corral et al. [13] introduced a geometry-based contact model that considers penetration for evaluating dis-

placement and restitution. One interesting feature of this model is that it formulates infinite planes to identify the contact points; however, this model does not combine well with differential algebraic equations (DAEs) or ordinary differential equations (ODEs) for constrained multibody systems.

Dopico et al. [14, 15] and Choi et al. [16] proposed a contact model that can be incorporated in index-3 DAE models. This approach uses a discretized mesh representation of geometry to identify the contact points. Therefore, the resulting constraint manifold is defined with equations corresponding to each node. These methods account for contact dynamics by forcing penalty factors using an augmented Lagrangian method. The equations of motion at the contact point are solved by projection of velocity and acceleration at contact points to solve for static equilibrium.

The contact model developed by Nikravesh and Lankarani [17] has been widely adopted, including in commercial software packages. This model uses a combination of nonlinear spring and damping elements to simulate the unilateral contact. Using this model, Shen et al. [18] developed a formulation for simulating impact dynamics by using dynamic optimization with energy loss being the objective function. A drawback of this contact model is that high stiffness values of the nonlinear springs result in "stiff" differential equations, which require very small time step sizes for numerical integration. Furthermore, separate detection of each contact and subsequent evaluation of contact forces makes the approach computationally inefficient, especially when multiple contacts occur intermittently. Using interpenetration for evaluating contact forces violates the constraints manifold.

Mechanical joints with clearances pose a special condition of "small gap systems" where contact formation and breaking-off can happen at high frequencies. Dynamics with clearance in joints was studied in Sharf and Zhang [19] and [20]. Sharf and Zhang in [19] founded the geometry-based contact models. Zakhariev [20] described an approach for calculating the reaction and friction forces between pairs of bodies in spatial mechanical systems as well as contact points in joints with clearances. Matrix methods were used to derive the nonlinear kinematic constraint equations, and the external and inertia forces for each configuration of the kinematic chain the contact points and corresponding normal forces were calculated. A drawback of the methodology in [20] is that the derived equality constraints do not account for all the possible orientations of the contact formation.

Further developments regarding joints with clearance include the work by Ibrahimi et al. [21], Flores and Lankarani [22], and Xiang et al. [23], demonstrated through a number of simulations on planar systems formulated as DAEs and contact model referenced from [17]. Time independent sensitivity analysis for planar systems was also included in [21]. Bauchau et al. [24] used unilateral contact conditions to simulate planar and spatial joints by adding additional rotational state variables and using nonholonomic inequality constraints with introduction of slack variables.

Another approach to deal with contact problems is the nonsmooth dynamics method. Moreau formulated the noncontinuous properties of nonsmooth mechanics [25–27]. Panagiotopoulos studied problems involving inequalities [28] and developed the formulations for hemivariational inequalities for mechanics, robotics, and neural network problems [29]. Moreover, the book [30] illustrates a proof about the validity of the impact theory given by Jean and Moreau. Since 1970s and 1980s, the theoretical and the mathematical formulation have substantially progressed, and have been demonstrated in industrial applications. Further, a comprehensive textbook by Acary and Brogliato [31, 32] covers a wide range of applications in mechanical and electrical systems. The book by Leine [33] investigates bifurcations of equilibrium in nonsmooth systems. Furthermore, Jean and Moreau

[34] contributed substantially to modeling dry friction and frictional impacts. This topic is covered in great detail by Brogliato [35]. The evolution of such processes has been covered in great detail in the literature such as Fremond [36], Glocker [37], Pfeiffer [38], and Wriggers [39].

Solving nonsmooth problems requires effective numerical solution strategies [31, 40]. The problem for planar nonsmooth systems results into a linear complementarity problem (LCP) and for spatial systems a nonlinear complementarity problem (NCP). The extended linear programming theories assure the existence of a solution for LCPs. For NCPs various formulations have been suggested and demonstrated through case-studies [41–43]. Further, Cosimo et al. [44, 45] gave efficient numerical algorithms based on decoupling methods applied with α -integration schemes. The comprehensive work by Brogliato et al. [46] includes various numerical methods for solving nonsmooth problems. Moreover, one of the most efficient theories of multibody systems with contacts, as given by Bremer [47] and Bremer and Pfeiffer [48], essentially applies projection methods on multibody dynamics theory. With time, this type of multibody theory has been combined with inequality formulations to describe the unilateral behavior of multiple contacts [49–51]. The applicability of this theory to continuous and discontinuous systems, as well as impact dynamics, makes it a versatile formulation [52–54].

Recently, the effectiveness of nonsmooth dynamics approaches has been demonstrated in many-body dynamics problems such as simulation of sand particles, stacking of bricks, and stacking of multiple balls in a box [55-57]; in all these applications the formulation of constraints is straightforward. When implementing a nonsmooth dynamics approach to multibody systems, a crucial aspect is the formulation of constraints. Because of shape and geometry of the components, the constraint formulation should capture the possibilities of multiple contact points at different locations while retaining the analytical nature of the constraint expression. Negrut et al. [58] described a methodology through which this can be achieved. The methodology poses the dynamics of the rigid bodies as a nonlinear programming problem (NLP) and uses dynamic optimization for minimizing an objective function derived from augmented Lagrangian approach. The definitions of cylindrical and revolute clearance joints given by Orden [59], Ambrosio et al. [60, 61], Liu [62], and Marques and Flores [63–65] consider angular misalignment as an inequality constraint instead of constraining both the possible contact points individually. These definitions have widely been used in **rp** formulations with continuous contact models. However, in the **rA** formulation, the constraint on angular misalignment is not needed. This was discussed in detail in Section 3.1. Further, the discussed methodologies were demonstrated on spherical joints [66] and planar prismatic joints in [67], and more recently, on a spatial revolute joints in [68-70]. While the formulation explained in [68] uses a similar approach to define revolute joints with clearance as discussed in this paper, it only considers radial clearance in the definition of joint and neglects axial clearance. On the other hand, Akhadkar et al. [69, 70] used commercial software packages for contact detection, and the formulation of joints was not covered by the authors. Further, planar prismatic joints were studied in detail by Flores et al. [67, 71]. However, the spatial prismatic joints have not received much attention, and hence the requirement of a robust definition of a spatial prismatic joint needs to be realized.

Studying a vast background of the literature, a need is thus realized to provide the general reader a framework for computer implementation of nonsmooth methods on multibody systems with clearances. This paper provides the reader a nonsmooth dynamics based framework dedicated to the simulation of multibody systems having frictionless ideal joints and joints with clearances. Specific contributions of this paper are as follows:

- Analytical constraint expressions are derived for several fundamental joints with clearances: cylindrical, revolute, and prismatic. Key differences between the proposed formulation of cylindrical and revolute joints and their precursors are discussed.
- 2. Not much work has been previously devoted to spatial prismatic joints with clearance. Here, a detailed definition of spatial prismatic joint with clearances is included.
- 3. A general NLP formulation is given for discretizing multibody dynamics of systems encompassing these fundamental joints with clearances.
- 4. An adaptive time-stepping scheme is discussed for the discrete NLP time integration.
- 5. The versatility of the framework is demonstrated through numerical experiments.

The remainder of this paper is structured as follows. Section 2 reviews the general NLP formulation applied to the time discretization of multibody systems. Section 3 formulates constraint expressions for spatial joints with clearances as functions of the rigid body pair's geometric features. Section 4 presents the case studies considered for testing the proposed formulation. Section 5 reports numerical results that illustrate the capabilities of the proposed approach and highlight the different dynamical behavior observed when joint clearances are accounted for. Section 6 concludes with key observations and a discussion on further scope of the work.

2 Methodology

2.1 Multibody dynamics in NLP formulation

This section illustrates the formulation of multibody systems in the general form of a nonlinear programming problem. Subsection 2.1.1 describes the equations of motion for rigid bodies undergoing motion in the absence of any constraints. Subsection 2.1.2 illustrates a method to evaluate contact forces on a set of rigid bodies. Subsection 2.1.3 combines the elements of free body dynamics and contact forces into an NLP of a general form. Finally, in Subsection 2.1.4, strategies to solve the general NLP problem through temporal discretization are discussed.

2.1.1 The free body dynamics

Formulation of equations of motion for three-dimensional systems in the body-frame coordinates results in a constant full rank and invertible 6×6 mass matrix for each body [72–74]. Considering these essential requirements, this section describes the free body dynamics.

Consider a rigid body of mass *m*, inertia matrix **J**, generalized coordinates (placed at the center of mass) $\mathbf{q} = [\mathbf{r}^T, \mathbf{p}^T]^T \in \mathbb{R}^7$ in global frame, and the corresponding time derivatives of generalized coordinates $\dot{\mathbf{q}} = [\dot{\mathbf{r}}^T, \dot{\mathbf{p}}^T]^T \in \mathbb{R}^7$, where $\mathbf{r} \in \mathbb{R}^3$ is the set of Cartesian [x, y, z] coordinates and $\mathbf{p} \in \mathbb{R}^4$ is the set of Euler parameters $[e_0, \mathbf{e}^T]^T$, all in global frame of reference. The equations of motion from D'Alembert's principle [75] are as follows:

$$\delta \mathbf{r}^{T} (m \, \ddot{\mathbf{r}} - \mathbf{F}_{\mathbf{A}}) + \delta \pi'^{T} \left(\mathbf{J} \, \dot{\boldsymbol{\omega}}' + \tilde{\boldsymbol{\omega}}' \, \mathbf{J} \, \boldsymbol{\omega}' - \boldsymbol{\tau}'_{\mathbf{A}} \right) = 0, \tag{4}$$

where $\delta \mathbf{r} \in \mathbb{R}^3$ are the infinitesimally small virtual displacements in global coordinate system and $\delta \pi' \in \mathbb{R}^3$ and virtual rotations in body-fixed coordinate system. The operator '.'

gives out the skew-symmetric matrix of the body-fixed vector of angular velocity $\boldsymbol{\omega}'$. The vector $\mathbf{F}_{\mathbf{A}} \in \mathbb{R}^3$ represents externally applied forces in global reference frame, and $\boldsymbol{\tau}'_{\mathbf{A}} \in \mathbb{R}^3$ are the externally applied torques in body-fixed reference frames. Equation (4) can thus be decomposed into the following set of equations in the absolute coordinate system, also called **rA** coordinate formulation [76, 77]:

$$m \ddot{\mathbf{r}} = \mathbf{F}_{\mathbf{A}},\tag{5a}$$

$$\mathbf{J}\,\dot{\boldsymbol{\omega}}' + \tilde{\boldsymbol{\omega}}'\,\mathbf{J}\,\boldsymbol{\omega}' = \boldsymbol{\tau}'_{\mathbf{A}}.\tag{5b}$$

Equations (5a) and (5b) can be written in a matrix form as follows:

$$\begin{bmatrix} m\mathbf{I} & \mathbf{0} \\ \mathbf{0} & \mathbf{J} \end{bmatrix} \begin{bmatrix} \ddot{\mathbf{r}} \\ \dot{\boldsymbol{\omega}}' \end{bmatrix} + \begin{bmatrix} \mathbf{0} \\ \tilde{\boldsymbol{\omega}}' \mathbf{J} \, \boldsymbol{\omega}' \end{bmatrix} - \begin{bmatrix} \mathbf{F}_{\mathbf{A}} \\ \boldsymbol{\tau}'_{\mathbf{A}} \end{bmatrix} = \mathbf{0}.$$
 (6)

Similarly, for a system of N bodies, equation (6) takes the form

$$\begin{bmatrix} m_{1}\mathbf{I} & \mathbf{0} & \mathbf{0} & \mathbf{0} & \mathbf{0} & \dots & \mathbf{0} \\ \mathbf{0} & \mathbf{J}_{1} & \mathbf{0} & \mathbf{0} & \mathbf{0} & \dots & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & m_{2}\mathbf{I} & \mathbf{0} & \mathbf{0} & \dots & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \mathbf{0} & \mathbf{J}_{2} & \mathbf{0} & \dots & \mathbf{0} \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ \mathbf{0} & \mathbf{0} & \mathbf{0} & \mathbf{0} & \dots & \mathbf{m}_{N}\mathbf{I} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \mathbf{0} & \mathbf{0} & \dots & \mathbf{0} & \mathbf{J}_{N} \end{bmatrix} \begin{bmatrix} \ddot{\mathbf{r}}_{1} \\ \dot{\boldsymbol{\omega}}_{1}' \\ \ddot{\mathbf{r}}_{2} \\ \dot{\boldsymbol{\omega}}_{2}' \\ \vdots \\ \vdots \\ \ddot{\mathbf{r}}_{N} \\ \dot{\boldsymbol{\omega}}_{N}' \end{bmatrix} + \begin{bmatrix} \mathbf{0}_{1} \\ \tilde{\boldsymbol{\omega}}_{1}' \mathbf{J}_{1} \boldsymbol{\omega}_{1} \\ \mathbf{0}_{2} \\ \vdots \\ \tilde{\boldsymbol{\omega}}_{2}' \mathbf{J}_{2} \boldsymbol{\omega}_{2}' \\ \vdots \\ \mathbf{0}_{N} \\ \tilde{\boldsymbol{\omega}}_{N}' \mathbf{J}_{N} \boldsymbol{\omega}_{N}' \end{bmatrix} - \begin{bmatrix} \mathbf{F}_{A1} \\ \mathbf{\tau}_{A1}' \\ \mathbf{F}_{A2} \\ \mathbf{\tau}_{A2}' \\ \vdots \\ \mathbf{F}_{AN} \\ \mathbf{\tau}_{AN}' \end{bmatrix} = \mathbf{0}, \quad (7)$$

which in a matrix form reads

$$\mathbf{M}\,\dot{\mathbf{v}} + \mathbf{S}\,(\mathbf{q},\,\dot{\mathbf{q}}) - \mathbf{Q}_{\mathbf{A}} = \mathbf{0}.\tag{8}$$

In (8), $\dot{\mathbf{v}} \in \mathbb{R}^{6N}$ is the vector of time derivatives of the body-fixed velocity vector $\mathbf{v} \in \mathbb{R}^{6N}$ for *N* bodies. The matrix $\mathbf{M} \in \mathbb{R}^{6N \times 6N}$ combined mass matrix of *N* bodies in the absolute coordinate system. The vector $\mathbf{S}(\mathbf{q}, \dot{\mathbf{q}}) \in \mathbb{R}^{6N}$ represents the vector of forces because of rotation, and the vector $\mathbf{Q}_{\mathbf{A}} \in \mathbb{R}^{6N}$ represents external forces (in global frame) and torque (in body fixed coordinate system) acting on the body neglecting friction. The mapping between the body-fixed angular velocity vector $\boldsymbol{\omega}'$ and the time derivatives of Euler parameters $\dot{\mathbf{p}}$ is given using the matrix $\mathbf{G}(\mathbf{p})$ composed of orthogonal column vectors as follows [75]:

$$\mathbf{G} (\mathbf{p}) := \begin{bmatrix} -\mathbf{e} & -\tilde{\mathbf{e}} + e_0 \mathbf{I} \end{bmatrix},$$

$$\dot{\mathbf{p}} = \frac{1}{2} \mathbf{G}^T (\mathbf{p}) \boldsymbol{\omega}',$$

$$\boldsymbol{\omega}' = 2 \mathbf{G} (\mathbf{p}) \dot{\mathbf{p}}.$$

(9)

Unless the contact happens, the bodies move freely as an unconstrained system and have the unconstrained accelerations:

$$\dot{\mathbf{v}}_{uc} = -\mathbf{M}^{-1} \left(\mathbf{S} \left(\mathbf{q}, \dot{\mathbf{q}} \right) - \mathbf{Q}_{\mathbf{A}} \right). \tag{10}$$

2.1.2 Dynamics with contact

Consider two bodies A and B in contact, as shown in Figure 1. Their respective centers of mass are c_A and c_B , and their coordinates in the absolute coordinate system are \mathbf{r}_A and \mathbf{r}_B ,



Fig. 1 Schematic representation of bodies in contact (Color figure online)

respectively. The contact point is detected at a location indicated by body fixed vectors $\mathbf{s'}_A$ and $\mathbf{s'}_B$ in bodies *A* and *B* respectively. Let \mathbf{n}_j be the normal unit vector along which the contact force acts on body *A* at the *j*th contact in the global reference frame. This is required because all the forces are evaluated in the global reference frame [58]. The corresponding unit vector for body *B* will be $-\mathbf{n}_j$. If the magnitude of contact force at the *j*th contact is denoted by λ_j , the combined vector of contact forces in the global reference frame and the torques generated by them in body-fixed frame can be evaluated by equation (11), as derived in [58]:

$$\mathbf{F}_{\mathbf{C}j} = \begin{bmatrix} \mathbf{n}_j \\ \tilde{\mathbf{s}}'_A \mathbf{A}^T(\mathbf{p}_A) \mathbf{n}_j \\ -\mathbf{n}_j \\ -\tilde{\mathbf{s}}'_B \mathbf{A}^T(\mathbf{p}_B) \mathbf{n}_j \end{bmatrix} \lambda_j = \mathbf{D}_j \lambda_j, \qquad (11)$$

where $\mathbf{A} \in \mathbb{R}^{3\times3}$ is the rotation matrix as a function of the Euler parameters of bodies *A* and *B*, $\mathbf{D}_j \in \mathbb{R}^{6N\times1}$ is the contact matrix, and the operator $\ddot{\cdot}$ gives out the skew-symmetric matrix of the underlying arbitrary vector. Similarly, for a number of possible *M* contacts between the two bodies, the total contact force $\mathbf{F}_{\mathbf{C}}$ is the sum of all the contact forces acting on the respective bodies, where $\mathbf{D} \in \mathbb{R}^{6N\times M}$ and $\lambda_{\mathbf{c}} \in \mathbb{R}^{M\times1}$, and is given as follows:

$$\mathbf{F}_{\mathbf{C}} = \begin{bmatrix} \mathbf{D}_1 & \mathbf{D}_2 & \dots & \mathbf{D}_M \end{bmatrix} \begin{bmatrix} \lambda_1 \\ \lambda_2 \\ \vdots \\ \lambda_M \end{bmatrix} = \mathbf{D}\lambda_{\mathbf{c}}.$$
 (12)

Using the same construct, the relative normal velocity between the two bodies at the contact point j in the mixed frame of reference is [58]

$$\mathbf{v}_{\mathbf{rel}\,j} = \mathbf{D}_{j}^{T} \begin{bmatrix} \dot{\mathbf{r}}_{A} \\ \boldsymbol{\omega}'_{A} \\ \dot{\mathbf{r}}_{B} \\ \boldsymbol{\omega}'_{B} \end{bmatrix} = \mathbf{D}_{j}^{T} \mathbf{v}.$$
(13)

Deringer

2.1.3 The nonlinear programming problem

Equation (10) is an expression to evaluate the unconstrained accelerations in the absolute coordinates. In addition, the orthonormality of Euler parameters has to be enforced for each body as a bilateral equality constraint [75]:

$$\mathbf{B}(\mathbf{q}) = \begin{bmatrix} \mathbf{p}_1^T \, \mathbf{p}_1 - 1 \\ \mathbf{p}_2^T \, \mathbf{p}_2 - 1 \\ \vdots \\ \mathbf{p}_N^T \, \mathbf{p}_N - 1 \end{bmatrix} = \mathbf{0}.$$
 (14)

Therefore, there exists a set of Lagrange multipliers $\lambda_{eq} \in \mathbb{R}^{N \times 1}$ such that

$$\mathbf{M}\,\dot{\mathbf{v}} + \mathbf{B}_{\mathbf{q}}^{T}\left(\mathbf{q}\right)\,\boldsymbol{\lambda}_{\mathbf{eq}} + \mathbf{S}(\mathbf{q},\,\dot{\mathbf{q}}) - \mathbf{Q}_{\mathbf{A}} = \mathbf{0},\tag{15}$$

where $\mathbf{B}_{\mathbf{q}}(\mathbf{q})$ is the Jacobian matrix of $\mathbf{B}(\mathbf{q})$ in equation (14). The state-of-the-art nonsmooth method proposed by [58] has been demonstrated on granular materials of spherical structure. Such particles have minute inertia and are uniformly distributed because of the spherical shape. In such cases, rotation does not impact the state variables to a significant extent, and hence (14) can be neglected for such cases. However, for rigid bodies, the inertia is significant and is often not uniformly distributed. Indeed, the rotation of the component can affect the stability of the solution [78]. Hence, in cases of rigid bodies, the normalization constraint on Euler parameters needs to be enforced to ensure orthogonality of basis vectors of the body-fixed frame.

Further, let the gap function $\Phi^{\text{ineq}} \in \mathbb{R}^{M \times 1}$ represent the vector of algebraic inequalities for M unilateral contacts in the system. Let the condition of nonzero gap be such that $\Phi_j^{\text{ineq}} \leq 0, j \in 1, 2, ..., M$, where each element j of Φ^{ineq} has a negative value. If the contact is formed, i.e., $\Phi_j^{\text{ineq}} = 0, j \in 1, 2, ..., M$, the addition of the contact forces thus generated $D\lambda_c$ results in constrained equation of motion as follows, where $\dot{\mathbf{v}}$ is the vector of constrained accelerations:

$$\dot{\mathbf{v}} = -\mathbf{M}^{-1} \left(\mathbf{S} \left(\mathbf{q}, \dot{\mathbf{q}} \right) + \mathbf{B}_{\mathbf{q}}^{T} (\mathbf{q}) \, \boldsymbol{\lambda}_{\mathbf{eq}} + \mathbf{D}(\mathbf{q}) \, \boldsymbol{\lambda}_{\mathbf{c}} - \mathbf{Q}_{\mathbf{A}} \right), \tag{16a}$$

$$\Phi^{\text{ineq}}(\mathbf{q}) * \lambda_{\mathbf{c}} = \mathbf{0}; \lambda_{\mathbf{c}} \ge \mathbf{0}, \tag{16b}$$

where the notation $\Phi^{\text{ineq}}(\mathbf{q}) * \lambda_{\mathbf{c}}$ represents the element-wise product of every expression in the constraint manifold and the corresponding magnitude of contact force. This complementarity condition states that when there is no contact, the *j*th constraint expression $\Phi_j^{\text{ineq}} < 0$ and the contact force's magnitude $\lambda_{\mathbf{c}j} = 0$. However, when the contact happens, $\Phi_j^{\text{ineq}} = 0$ and the contact force's magnitude $\lambda_{\mathbf{c}j} > 0$.

Furthermore, Gauss' principle of least constraint [79, 80] states that the true acceleration of a mechanical system of N particulate masses $\mathbf{\ddot{r}}_j \in \mathbb{R}^3$ is the minimum of the following quantity:

$$\mathcal{J} = \sum_{j=1}^{N} m_j \left\| \ddot{\mathbf{r}}_j - \frac{\mathbf{F}_j}{m_j} \right\|^2 \quad \forall j \in [1, 2, 3, ..., N],$$
(17)

where the j^{th} particle has mass m_j , position vector $\mathbf{r}_j \in \mathbb{R}^3$, and applied vector of nonconstraint force $\mathbf{F}_j \in \mathbb{R}^3$ acting on the mass. Equation (17) is also applicable to rotational ac-

celerations as shown by Udwadia and Yang et al. [79, 80] and Glocker [37]. With a constant mass matrix \mathbf{M} and plugging equations (10), (16a) and (16b) in equation (17), the governing objective function for the system can be represented as the minimum of the difference between constrained and unconstrained accelerations as shown in equation (18).

$$\mathcal{J} = (\dot{\mathbf{v}} - \dot{\mathbf{v}}_{uc})^T \mathbf{M} (\dot{\mathbf{v}} - \dot{\mathbf{v}}_{uc})$$

= $\left(\mathbf{M}^{-1} (\mathbf{D} \lambda_c + \mathbf{B}_q^T \lambda_{eq}) \right)^T \mathbf{M} \left(\mathbf{M}^{-1} (\mathbf{D} \lambda_c + \mathbf{B}_q^T \lambda_{eq}) \right).$ (18)

Since M is a constant symmetric matrix, equation (18) can be written as follows:

$$\mathcal{J} = \left(\mathbf{D}\boldsymbol{\lambda}_{\mathbf{c}} + \mathbf{B}_{\mathbf{q}}^{T}\boldsymbol{\lambda}_{\mathbf{eq}}\right)^{T} \mathbf{M}^{-1} \left(\mathbf{D}\boldsymbol{\lambda}_{\mathbf{c}} + \mathbf{B}_{\mathbf{q}}^{T}\boldsymbol{\lambda}_{\mathbf{eq}}\right).$$
(19)

It must be noticed that the objective function in (17) holds true for hyperstatic case because the contacting bodies are rigid. For simulating soft (flexible) contacts, an additional damping matrix gets added to the Schur complement $[\mathbf{D} \ \mathbf{B}_{\mathbf{q}}^T]^T \mathbf{M}^{-1}[\mathbf{D} \ \mathbf{B}_{\mathbf{q}}^T]$, as described in [81]. Further, flexible systems often result in under-constrained mathematical models, and dealing with such systems is out of the scope of the work presented here. Hence the rigidbody dynamics problem can be stated as a nonlinear programming problem (20) for the vector of state variables $\mathbf{Y} := [\boldsymbol{\lambda}_c^T, \boldsymbol{\lambda}_{eq}^T, \dot{\mathbf{q}}^T, \mathbf{q}^T]^T$. Here the objective function (19) is minimized subject to the constraints imposed by equations of motion and the complementary constraints (16a) and (16b), and the normalized Euler parameters (14):

$$\begin{split} \hat{\mathbf{Y}} &= \arg\min_{\mathbf{Y}} \quad \left(\mathbf{D}\lambda_{\mathbf{c}} + \mathbf{B}_{\mathbf{q}}^{T}\lambda_{\mathbf{eq}}\right)^{T} \mathbf{M}^{-1} \left(\mathbf{D}\lambda_{\mathbf{c}} + \mathbf{B}_{\mathbf{q}}^{T}\lambda_{\mathbf{eq}}\right) \\ \text{s.t.} \quad \mathbf{M}\dot{\mathbf{v}} + \mathbf{B}_{\mathbf{q}}^{T}(\mathbf{q})\lambda_{\mathbf{eq}} + \mathbf{D}(\mathbf{q})\lambda_{\mathbf{c}} + \mathbf{S}\left(\mathbf{q},\dot{\mathbf{q}}\right) - \mathbf{Q}_{\mathbf{A}} = \mathbf{0}, \\ \mathbf{B}(\mathbf{q}) &= 0, \\ \Phi^{\text{ineq}}\left(\mathbf{q}\right) \leq \mathbf{0} \perp \lambda_{\mathbf{c}} \geq \mathbf{0}. \end{split}$$
(20)

Furthermore, to simulate a multibody system with a combination of equality type constraints $\Phi^{eq} = \mathbf{0}$, i.e., for ideal joints, the manifold Φ^{eq} can be appended to the bilateral constraint manifold (14), as discussed in Section 2.2. The capacity of the described NLP formulation to deal with ideal as well as contact-driven constraints makes it an all-inclusive formulation.

2.1.4 Discretization and solution strategy

Consider now the discrete version of the dynamics. At time t_{n-1} one has available the global frame time derivatives of generalized coordinates $\dot{\mathbf{q}}_{n-1}$, the velocities \mathbf{v}_{n-1} , and the positions of the generalized coordinates \mathbf{q}_{n-1} as follows:

$$\mathbf{q}_{n-1} = \begin{bmatrix} \mathbf{r}_{n-1}^T & \mathbf{p}_{n-1}^T \end{bmatrix}^T,$$
(21a)

$$\mathbf{v}_{n-1} = \begin{bmatrix} \dot{\mathbf{r}}_{n-1}^T & (2 \mathbf{G} (\mathbf{p}_{n-1}) \dot{\mathbf{p}}_{n-1})^T \end{bmatrix}^T, \qquad (21b)$$

$$\dot{\mathbf{q}}_{n-1} = \begin{bmatrix} \dot{\mathbf{r}}_{n-1}^T & \dot{\mathbf{p}}_{n-1}^T \end{bmatrix}^T.$$
(21c)

Let h_n be the time step size. Then, at the new derivatives of generalized coordinates $\dot{\mathbf{q}}_n$, the velocities \mathbf{v}_n and the positions of the generalized coordinates \mathbf{q}_n must be evaluated at

time $t_n = t_{n-1} + h_n$. Hence, using half implicit Euler's expansion [82], we have the following relationships:

$$\dot{\mathbf{q}}_n = \begin{bmatrix} \dot{\mathbf{r}}_n^T & \dot{\mathbf{p}}_n^T \end{bmatrix}^T, \qquad (22a)$$

$$\mathbf{q}_n = \mathbf{q}_{n-1} + h_n \dot{\mathbf{q}}_n \tag{22b}$$

$$\mathbf{v}_n = \begin{bmatrix} \dot{\mathbf{r}}_n^T & (2 \mathbf{G} (\mathbf{p}_{n-1}) \dot{\mathbf{p}}_n)^T \end{bmatrix}^T, \qquad (22c)$$

$$\dot{\mathbf{v}}_n = \frac{\mathbf{v}_n - \mathbf{v}_{n-1}}{h_n},\tag{22d}$$

$$t_n = t_{n-1} + h_n.$$
 (22e)

The bilateral constraint expressions (14) can be further discretized using Taylor's expansion. The unilateral constraints can be discretized using the relative velocities in equation (13). These discretizations are represented in the following equations:

$$\mathbf{B}(\mathbf{q}_n) \approx \mathbf{B}(\mathbf{q}_{n-1}) + h_n \, \mathbf{B}_{\mathbf{q}}(\mathbf{q}_{n-1}) \, \dot{\mathbf{q}}_n = \mathbf{0}, \tag{23}$$

$$\mathbf{\Phi}^{\text{ineq}}(\mathbf{q}_n) \approx \mathbf{\Phi}^{\text{ineq}}(\mathbf{q}_{n-1}) + h_n \mathbf{D}^T(\mathbf{q}_{n-1}) \mathbf{v}_n \le \mathbf{0}.$$
 (24)

Incorporating equations (21a)–(21c), (22a)–(22e), (23), and (24) in NLP (20) we get the following NLP in discretized form, with respect to the vector of optimization variables $[h_n, \mathbf{Y}_n]$:

$$\begin{bmatrix} \hat{h}_n, \hat{\mathbf{Y}}_n \end{bmatrix} = \arg\min_{h_n, \mathbf{Y}_n} \quad \mathbf{T}_n^T \mathbf{M}^{-1} \mathbf{T}_n$$

s.t.
$$\mathbf{M}(\mathbf{v}_n - \mathbf{v}_{n-1}) + h_n \mathbf{K}_n = \mathbf{0},$$
$$\mathbf{B}(\mathbf{q}_{n-1}) + h_n \mathbf{B}_{\mathbf{q}}(\mathbf{q}_{n-1}) \dot{\mathbf{q}}_n = \mathbf{0},$$
$$\mathbf{\Phi}^{\text{ineq}}(\mathbf{q}_{n-1}) + h_n \mathbf{D}_{n-1}^T \mathbf{v}_n \le \mathbf{0},$$
$$\lambda_{\mathbf{c}n} \ge \mathbf{0},$$
$$h_{\text{max}} \ge h_n \ge h_{\text{min}},$$
$$(25)$$

where \hat{h}_n and $\hat{\mathbf{Y}}_n$ are the resulting step size and solution, respectively, and \mathbf{T}_n and \mathbf{K}_n are defined as follows:

$$\mathbf{T}_{n} := \mathbf{D}(\mathbf{q}_{n-1}) \,\boldsymbol{\lambda}_{\mathbf{c}n} + \mathbf{B}_{\mathbf{q}}^{T}(\mathbf{q}_{n-1}) \,\boldsymbol{\lambda}_{\mathbf{eq}_{n}}, \tag{26a}$$

$$\mathbf{K}_n := \mathbf{T}_n + \mathbf{S} \; (\mathbf{q}_{n-1}, \dot{\mathbf{q}}_{n-1}) - \mathbf{Q}_{\mathbf{A}}(\mathbf{q}_{n-1}). \tag{26b}$$

Here, the optimal values of h_n , λ_{eq_n} , λ_{eq_n} , and $\dot{\mathbf{q}}_n$ need to be found at time t_n . The scalar terms h_{\min} and h_{\max} respectively are the minimum and maximum finite positive values of time step size. Once the optimization problem (25) has been solved at time t_n , the values of $\dot{\mathbf{q}}_n$ and h_n can then be utilized to evaluate and store the accelerations $\dot{\mathbf{v}}_n$ and the positions \mathbf{q}_n using equations (22a)–(22e).

It can be noticed that time step size has been considered as a variable parameter bounded by h_{\min} and h_{\max} respectively. This time-stepping scheme is purely based on enforcing the feasibility of both: discretized inequality (complementary) constraints representing contact



Fig. 2 Visual representation of step size variation (Color figure online)

and equality constraints that include equations of motion and discretized equality constraints velocities. Figure 2 shows the feasible region of inequality and equality constraints sets represented by $\Phi_n \leq 0$ and $B_n = 0$ respectively. Equation (22a)–(22e) gives the integrand \mathbf{q}_n at n^{th} time step. The step size h, essentially, defines the fraction of the length of the vector $\dot{\mathbf{q}}$. Further, there could be a situation with fixed step size such that the next step solution lies out of the feasible region hence resulting in violation of constraints. The two plots in Figure 2 represent such a general situation. In the left plot, h_{opt} indicates the optimal value of h as per the condition imposed by the linearized inequality constraint. In the right plot, however, a situation is shown where the step size h_{opt} violates the linearized equality constraints; in such a condition, h_{min} will be chosen by the algorithm. If the vector $\dot{\mathbf{q}}$ is bound to attain a certain magnitude for convergence, then the bounded variable h provides an extra degree of freedom and helps attain convergence. However, it also results in increase in dimensionality of the problem.

It must be mentioned here that the approach described in this section inherently assumes an inelastic nature of the contact, which is suitable for the scope of this work. The clearances are very small compared to the other dimensions of the components in a machine, and hence the components do not gain enough pre-collision velocities for the contact to fall in the elastic domain.

2.2 Implementation of frictionless ideal joints using the nonsmooth framework

Consider two bodies *A* and *B* in constrained to each other between points *P* and *Q* by a set of algebraic equations $\Phi(\mathbf{q}_A, \mathbf{q}_B) = \mathbf{0}$. Their respective centers of mass Cartesian coordinates in the global coordinate system are \mathbf{r}_A and \mathbf{r}_B , respectively. With a set of algebraic constraints defined for the k^{th} body in the system of *N* bodies with *M* total constraint equations, the constraint forces \mathbf{F}^k and torques \mathbf{T}'^k acting on the bodies can be evaluated in absolute coordinates using the approach discussed in [77], as presented in equation (27):

$$\mathbf{Q}^{k}_{const} = \begin{bmatrix} \mathbf{F}^{k} \\ \mathbf{T}^{\prime k} \end{bmatrix}_{const} = \begin{bmatrix} \mathbf{\Phi}^{T}_{\mathbf{r}} \\ \Pi^{\prime T}(\mathbf{\Phi}) \end{bmatrix}_{k} \boldsymbol{\lambda}_{\mathbf{c}k} = \mathbf{C}_{k} \boldsymbol{\lambda}_{\mathbf{c}k}, \qquad (27)$$

where λ_k is the Lagrangian multiplier associated with the k^{th} body equation $\Phi_k = 0$ and Φ_r is the Jacobian matrix of Φ with respect to Cartesian coordinates $\mathbf{r} \in [\mathbf{r}_A, \mathbf{r}_B, \dots, \mathbf{r}_N]$.

The operator $\Pi' \in [\Pi'_A, \Pi'_B, ..., \Pi'_N]$ on the constraint describes the first order variation of the joint frame located by **s'** in the body frame. In addition to this, the reference [77] also describes the method to formulate following fundamental constraints. Similarly, the vector \mathbf{Q}_c comprising the constraint forces and torques for a system of *N* bodies can be defined as in equation (28), where λ_c is the vector of all λ_{ck} .

$$\mathbf{Q}_{c} = \begin{bmatrix} \mathbf{C}_{1} & & \\ & \mathbf{C}_{2} & \\ & & \ddots & \\ & & & \mathbf{C}_{N} \end{bmatrix} \boldsymbol{\lambda}_{\mathbf{c}} = \mathbf{C}(\mathbf{q})\boldsymbol{\lambda}_{\mathbf{c}}.$$
(28)

Hence, using the methods for deriving NLP (20), the equations of motion for a smooth system using nonsmooth framework can be stated as NLP (29) for the vector of state variables $\mathbf{Y} := [\boldsymbol{\lambda}_{\mathbf{c}}^{T}, \dot{\mathbf{q}}^{T}, \mathbf{q}^{T}]^{T}$:

$$\begin{split} \hat{\mathbf{Y}} &= \arg\min_{\mathbf{Y}} \quad \mathbf{H}(\mathbf{q}, \lambda_{\mathbf{c}}, \lambda_{eq}) \, \mathbf{M}^{-1} \, \mathbf{H}(\mathbf{q}, \lambda_{\mathbf{c}}, \lambda_{eq}) \\ \text{s.t.} \quad \mathbf{M} \, \dot{\mathbf{v}} + \mathbf{H}(\mathbf{q}, \lambda_{\mathbf{c}}, \lambda_{eq}) + \mathbf{S} \, (\mathbf{q}, \dot{\mathbf{q}}) - \mathbf{Q}_{\mathbf{A}}(\mathbf{q}) = \mathbf{0}, \\ \mathbf{B}(\mathbf{q}) &= \mathbf{0}, \\ \Phi \left(\mathbf{q} \right) &= \mathbf{0}, \end{split}$$
(29)

where

$$\mathbf{H}(\mathbf{q}, \boldsymbol{\lambda}_{\mathbf{c}}, \boldsymbol{\lambda}_{\mathbf{eq}}) = \mathbf{B}_{\mathbf{q}}^{T}(\mathbf{q})\boldsymbol{\lambda}_{\mathbf{eq}} + \mathbf{C}(\mathbf{q})\boldsymbol{\lambda}_{\mathbf{c}}.$$
(30)

It can be noticed that the contact matrix **D** in NLP (20) is replaced by a mapping **C** and the inequality constraints vanish, leaving the rest of the structure unchanged. Using equations (21a)–(21c) and (22a)–(22e), the time-discretized equations for continuous systems can thus be written as follows with respect to the vector of optimization variables $\mathbf{Y}_n = [h_n, \lambda_{\mathbf{c}_n}^T, \lambda_{\mathbf{eq}_n}^T, \dot{\mathbf{q}}_n^T]^T$:

$$\hat{\mathbf{Y}}_{n} = \arg\min_{\mathbf{Y}_{n}} \quad \mathbf{H}_{n}^{T} \mathbf{M}^{-1} \mathbf{H}_{n}$$
s.t.
$$\mathbf{M}(\mathbf{v}_{n} - \mathbf{v}_{n-1}) + h_{n}(\mathbf{K}_{n}) = \mathbf{0},$$

$$\mathbf{B}(\mathbf{q}_{n-1}) + h_{n} \mathbf{B}_{\mathbf{q}}(\mathbf{q}_{n-1}) \dot{\mathbf{q}}_{n} = \mathbf{0},$$

$$\Phi(\mathbf{q}_{n-1}) + h_{n} \mathbf{C}_{n-1}^{T} \mathbf{v}_{n} = \mathbf{0},$$
(31)

where $\hat{\mathbf{Y}}_n$ is the resulting solution, and \mathbf{H}_n and \mathbf{K}_n are defined as follows:

$$\mathbf{H}_{n} := \mathbf{C}(\mathbf{q}_{n-1}) \,\boldsymbol{\lambda}_{\mathbf{c}n} + \mathbf{B}_{\mathbf{q}}^{T}(\mathbf{q}_{n-1}) \,\boldsymbol{\lambda}_{\mathbf{eq}_{n}}, \tag{32a}$$

$$\mathbf{K}_n := \mathbf{H}_n + \mathbf{S} \ (\mathbf{q}_{n-1}, \dot{\mathbf{q}}_{n-1}) - \mathbf{Q}_{\mathbf{A}}(\mathbf{q}_{n-1}). \tag{32b}$$

2.3 Algorithm

A step-by-step procedure to implement the formulation discussed in Sections 2.1 and 2.2. For a target simulation time t_{max} , the initial values of \mathbf{Y}_0 at time t_0 should be such that all

Algorithm 1 Nonsmooth multibody dynamics

| 1: | Inputs: t_0 , t_{final} , h_{\min} , h_{\max} , \mathbf{Y}_0 | The initial states $\mathbf{Y}_0 = [\boldsymbol{\lambda}_{\mathbf{c}_0}^T, \boldsymbol{\lambda}_{\mathbf{e}_0}^T, \dot{\mathbf{q}}_0^T]^T$ |
|-----|---|--|
| 2: | $n \leftarrow 0$ | |
| 3: | while $t_n < t_{\text{final}}$ do | |
| 4: | $n \leftarrow n+1$ | |
| 5: | $\begin{cases} \hat{h}_{\min} = \min\{h_{\min}, t_{\text{final}} - t_{n-1}\}\\ \hat{h}_{\max} = \min\{h_{\max}, t_{\text{final}} - t_{n-1}\}\end{cases}$ | \triangleright Ensure $t_n \leq t_{\text{final}}$ |
| 6: | $[h_n, \mathbf{Y}_n] \leftarrow \text{Solution of NLP}(\mathbf{Y}_{n-1}, \hat{h}_{\min}, \hat{h})$ | $(max) \qquad \qquad \triangleright \text{ Solve (25) or (31)}$ |
| 7: | $[\boldsymbol{\lambda}_{\mathbf{c}_n}^T, \boldsymbol{\lambda}_{\mathbf{e}\mathbf{q}_n}^T, \dot{\mathbf{q}}_n^T] := \mathbf{Y}_n^T$ | ⊳ Extract the state variables |
| 8: | $\begin{bmatrix} \mathbf{q}_n \\ \mathbf{v}_n \\ \dot{\mathbf{v}}_n \end{bmatrix} \leftarrow \begin{bmatrix} \mathbf{q}_{n-1} + h_n \dot{\mathbf{q}}_n \\ [\dot{\mathbf{r}}_n^T & (2 \mathbf{G} (\mathbf{p}_{n-1}) \dot{\mathbf{p}}_n)^T]^T \\ \frac{1}{h_n} (\mathbf{v}_n - \mathbf{v}_{n-1}) \end{bmatrix}$ | ⊳ from equation (22a)–(22e) |
| 9: | $t_n \leftarrow t_{n-1} + h_n$ | |
| 10: | end while | |

constraint equations are satisfied. For the initial values of generalized coordinates satisfying a negative value of gap function, the initial value of Lagrangian multipliers λ_{c0} should be assumed zero.

The values of state variables obtained at the n^{th} time-step in step 7 of the Algorithm 1 can be later used to evaluate contact forces and torques using equation (12).

3 Parametric definitions of joints with clearances

In this section, the formulations for three fundamental joints used in mechanical systems are presented. The gap function Φ^{ineq} should provide both the normal distance and the normal unit vector between the potential contact points to suit equation (11). Keeping this key requirement in consideration, the parametric formulations for joints with clearances are presented in this section.

3.1 Cylindrical joint with clearances

Figure 3 shows a system of two bodies *i* and *j* forming a cylindrical joint with radial clearance *c*. The generalized coordinates of bodies are $\mathbf{q}_i \in \mathbb{R}^7$ and $\mathbf{q}_j \in \mathbb{R}^7$ respectively. Let points P_{i_1} and P_{i_2} represent the two ends of center-line of body *i* in the global frame of reference. Let points P_{j_1} and P_{j_2} represent the two ends of center-line of body *j* in the global frame of reference. The perpendicular distances of points P_{i_1} and P_{i_2} from the center axis of body *j* are d_{s_1} and d_{s_2} , along with the unit vectors \mathbf{n}_{s_1} and \mathbf{n}_{s_2} , respectively. Let $\mathbf{A}(\mathbf{p}_i)$ and $\mathbf{A}(\mathbf{p}_j)$ be the rotation matrices of the respective bodies. Given the local coordinate vector \mathbf{s}'_{k_x} of an arbitrary point P_{k_x} with $x \in [1, 2]$ in a body with respect to the centroid $k \in [i, j]$, the position of point P_{k_x} can be evaluated in global frame as \mathbf{r}_{k_x} using equation (33).

$$\mathbf{r}_{k_x} = \mathbf{r}_k + \mathbf{A}(\mathbf{p}_k) \, \mathbf{s}'_{k_x}. \tag{33}$$

The body with the smaller axial length will have one or both of the end faces making contact with the longer body. Assuming that body *i* is the smaller body and considering P_{j_1} as the



Fig. 3 Schematic representation of a spatial cylindrical joint with clearance (Color figure online)

reference point with position vector \mathbf{r}_{j_1} , the perpendicular vectors \mathbf{d}_{s_x} , $x \in [1, 2]$ pointing towards the contact points P_{k_x} can be evaluated via the following equations:

$$\mathbf{d}_{s_x} = (\mathbf{r}_{i_x} - \mathbf{r}_{j_1}) - (\mathbf{r}_{i_x} - \mathbf{r}_{j_1}) \frac{(\mathbf{r}_{i_x} - \mathbf{r}_{j_1})^T (\mathbf{r}_{j_2} - \mathbf{r}_{j_1})}{\|\mathbf{r}_{i_x} - \mathbf{r}_{j_1}\| \|\mathbf{r}_{j_2} - \mathbf{r}_{j_1}\|},$$
(34)

$$\mathbf{n}_{s_x} = \frac{\mathbf{d}_{s_x}}{\|\mathbf{d}_{s_x}\|} = \frac{\mathbf{d}_{s_x}}{d_{s_x}}.$$
(35)

Considering c of a very small magnitude in comparison to other dimensions of the components, the perpendicular distances d_{s_1} and d_{s_2} must abide the following inequalities:

$$\mathbf{\Phi}^{\text{cylindric}}\left(\mathbf{q}\right) = \begin{bmatrix} d_{s_1} - c \\ d_{s_2} - c \end{bmatrix} \le \mathbf{0}.$$
(36)

Assuming the hole inside the body *i* has a radius *R*, the position of the corresponding contact points \mathbf{c}_x in the global reference frame can be calculated as follows:

$$\mathbf{c}_{x} = \mathbf{r}_{i} + \mathbf{A}(\mathbf{p}_{i})\mathbf{s}'_{i_{x}} - R\,\mathbf{n}_{s_{x}}.$$
(37)

It must be noticed that the definition of cylindrical and revolute clearance joints defined in [59–65] considers angular misalignment as an extra inequality constraint. These definitions have widely been used in **rp** formulations with continuous contact models. However, the requirement is to define the constraints for clearance joints in the **rA** formulation such that the normal vector at a contact point must be obtained as a function of the gap present between the colliding points, and the contact matrix thus formed must satisfy equation (24). For this requirement, it is essential that the inequality constraint set Φ must exclusively comprise distance quantities similar to what is defined in [68]. Hence, the cylindrical joint defined in equation (36) is $\Phi^{\text{cylindric}} \in \mathbb{R}^2$. Also, it can be noticed that the normal vector **n** required for



Fig. 4 Schematic representation of a spatial prismatic joint with clearance

evaluating **D** is a direct indicator of the relative normal direction at the contact. Hence, with the proposed definition, the joint is conducive to and can be integrated with the nonsmooth dynamics framework described in the previous sections. Since the function is evaluated in terms of absolute distance with $\lambda_c \geq 0$ and the normal vector is defined along the direction of the distance vector, the angular misalignment is automatically considered in this approach for all possible combinations of relative orientations of the bodies.

3.2 Prismatic joint with clearances

The spherical, revolute, and cylindrical joints with clearance [59–64, 66] and planar prismatic joints [67, 71] have been studied extensively. However, the spatial prismatic joints have not received much attention in the past. The prismatic joint with a clearance poses a special intricacy because of its exclusive dependence on geometric parameters. Traditionally, a prismatic joint consists of shapes with vertices and planes to constrain the rotational motion about the longitudinal axis, and these vertices form the contact points. A prismatic joint without clearance does not require these to be included as the contact forces are not evaluated on individual vertex. However, when clearance is included, each vertex of the inner body (minimum six are required for a spatial prismatic joint) needs to be individually constrained to not penetrate the enveloping body's two adjacent planes. Hence, the geometric locations of all the vertices are required with respect to the local coordinate system.

Figure 4 shows a system of two bodies *i* and *j* forming a prismatic joint with clearance. The generalized coordinates of the bodies are $\mathbf{q}_i \in \mathbb{R}^7$ and $\mathbf{q}_j \in \mathbb{R}^7$ respectively. Each vertex of body *i* (inner body) at front and back faces is constrained to two adjacent planes, as shown in Figure 5. For example, points P_{4f} and P_{4b} are constrained to planes 1 and 4, points P_{1f} and P_{1b} are constrained to plane 1 and plane 3, and so on. It requires at least three points to define a plane. The planes 1–4 are defined by the vertices in the above figure. For example, plane 1 is defined using points Q_{1f} , Q_{4f} , and Q_{4b} , plane 2 is defined by P_{1f} , P_{2f} , P_{1b} , and so on. Fig. 5 Each pair of vertices results in two values of directional distance vector (Color figure online)

With this methodology, a polygon of any regular or irregular shape can be defined with the help of all the vertices and planes. Each vertex of an N sided polygon is constrained to two adjacent planes at both the faces. This results in 4N equations of distance constraint. Obviously, each vertex's global coordinates need to be calculated as per equation (33). The perpendicular distance between a point P on body i, where $P \in [P_{1f}, P_{2f}, ..., P_{Nf}, P_{1b}, P_{2b}, ..., P_{Nb}]$, and a plane obtained from vertices on body j can be calculated as follows:

$$d_P = (\mathbf{r}_P - \mathbf{r}_Q)^T \mathbf{n}_{QP} \ge 0, \tag{38}$$

where \mathbf{n}_{QP} is the normal vector of the bounding plane pointing towards the point *P*, *Q* is any point in the bounding plane, and \mathbf{r}_P and \mathbf{r}_Q are the vectors associated with point *P* and point *Q* in the global reference frame evaluated with equation (33). Hence, for a joint of rigid bodies forming a prismatic joint, with the internal body having 2*M* vertices and bounded by external body having *M* planes, the constraint manifold is given as per the following inequality:

$$\boldsymbol{\Phi}^{\text{prismatic}}\left(\mathbf{q}\right) = \begin{bmatrix} -d_{P_{f1}} \\ \vdots \\ -d_{P_{f2M}} \\ -d_{P_{b1}} \\ \vdots \\ -d_{P_{b2M}} \end{bmatrix}_{4M \times 1} \leq \mathbf{0}.$$
(39)

Here, the subscripts 'f' and 'b' respectively signify the vertices in front and back, as shown in Figure 4. It can be inferred from equation (39) that the dimensionality of NLP (25) increases with the number of vertices in the joint. This is another justification for the lack of studies on prismatic joints with clearances, as compared to other fundamental joints.

3.3 Revolute joint with clearances

The spatial revolute joint is essentially an axially constrained cylindrical joint. Let body *i* in Figure 3 be axially constrained at the ends of body *j* such that there remains an axial clearance c_a . The normal vectors \mathbf{n}_{j_1} and \mathbf{n}_{j_2} at the constraining planes at P_{j_1} and P_{j_2} ,



| Table 1 Case study specifications: Cylindrical joint | Property | Value | Units |
|--|-----------------------|-------|-------|
| (Color figure online) | | | |
| | т | 3.2 | kg |
| | L | 100 | mm |
| | l | 20 | mm |
| | R | 5 | mm |
| | $\ \mathbf{s'}_{f}\ $ | 30 | mm |
| | F | 50 | Ν |

respectively, are evaluated as per the following equations:

$$\mathbf{n}_{j_1} = \frac{\mathbf{r}_{j_2} - \mathbf{r}_{j_1}}{\|\mathbf{r}_{j_2} - \mathbf{r}_{j_1}\|}, \qquad \mathbf{n}_{j_2} = -\mathbf{n}_{j_1}.$$
(40)

Consequently, the construction of the joint requires that the distance of points P_{i_1} and P_{i_2} from P_{j_1} and P_{j_2} , projected on the normal vectors \mathbf{n}_{j_1} and \mathbf{n}_{j_2} , respectively, is positive.

$$-(\mathbf{r}_{i_1} - \mathbf{r}_{j_1})^T \mathbf{n}_{j_1} \le 0,$$

$$-(\mathbf{r}_{i_2} - \mathbf{r}_{j_2})^T \mathbf{n}_{j_2} \le 0.$$
 (41)

Hence, the constraint manifold for a revolute joint with clearances is given as the following set of inequalities:

$$\boldsymbol{\Phi}^{\text{revolute}}\left(\mathbf{q}\right) = \begin{bmatrix} \boldsymbol{\Phi}^{\text{cylindric}}\left(\mathbf{q}\right) \\ -(\mathbf{r}_{i_1} - \mathbf{r}_{j_1})^T \mathbf{n}_{j_1} \\ -(\mathbf{r}_{i_2} - \mathbf{r}_{j_2})^T \mathbf{n}_{j_2} \end{bmatrix} \le \mathbf{0}.$$
(42)

4 Case studies

This section discusses the three relevant case studies that were used to test the methodology discussed in Sections 2 and 3.

4.1 Case study: cylindrical joint

Figure 6 shows a simple case study of a spatial cylindrical joint with clearance to illustrate the proof of concept for the proposed formulation. The case study comprises of a slider of mass *m* with its center of mass denoted by the generalized coordinates $\mathbf{q} \in \mathbb{R}^7$ and a massless rod of radius *R* fixed at both ends. There is a radial clearance c = 0.1 mm between the slider and the rod and gravity acts along the $-\mathbf{Y}$ axis. An external force *F* acts on the slider at a point marked by a body-frame vector $\mathbf{s'}_f$. The system specific details illustrated in Figure 6 are shown in Table 1.

4.2 Case study: prismatic joint

Figure 7 shows a simple case study of a spatial prismatic joint with clearance c = 0.3 mm. As it can be observed, the bodies have triangular cross-section with each edge of the cavity in body j of length e_j ; the corresponding length for body i is e_i . The slider i has mass



Fig. 7 Schematic diagram of the triangular prismatic joint with clearance (Color figure online)

m with its center of mass denoted by the generalized coordinates $\mathbf{q} \in \mathbb{R}^7$ and a mass-less triangular prism is fixed at both ends. The gravity acts along the $-\mathbf{Y}$ axis and an external force *F* acts on the slider at a point marked by a body-frame vector $\mathbf{s'}_f$. The topmost vertices of the slider are labeled P_{f_1} and P_{b_1} respectively. The trajectory of these two contact points is tracked and discussed in Section 5.2. The system specific details illustrated in Figure 7 are shown in Table 2.

4.3 Case study: revolute joint

Figure 8 shows a simple case study of a pendulum *i* forming a revolute joint with a hinge pin *j* of radius *R*. The radial and axial clearances are denoted by *c* and c_a . The center of mass of the pendulum is located at cm_x and cm_z units along local X and Z directions. As can be observed, the bob of the pendulum is a rod protruding out of the XZ plane. This odd shape has been chosen to induce motion along Z direction as well. The pendulum *i* has mass *m* with its center of mass denoted by the generalized coordinates $\mathbf{q} \in \mathbb{R}^7$ and the hinge pin *j*

| specifications: Prismatic joint | Property | Value | Units | |
|---------------------------------|-----------------------|-------|-------|--|
| (Color ligure olimic) | m | 3.2 | kg | |
| | L | 100 | mm | |
| | 1 | 20 | mm | |
| | ei | 17.67 | mm | |
| | ej | 17.32 | mm | |
| | $\ \mathbf{s'}_{f}\ $ | 30 | mm | |
| | F | 40 | Ν | |



Fig. 8 Schematic diagram of the revolute joint with clearance (Color figure online)

| Table 3 Case study specifications: Revolute joint | Property | Value | Units |
|---|----------|-------|-------|
| | m | 2 | kg |
| | l | 20 | m |
| | С | 0.1 | m |
| | R | 5 | m |
| | c_a | 0.1 | m |
| | cm_X | 30.9 | m |
| | cm_z | 11.97 | m |

is mass-less and fixed at both ends. The gravity acts along the -Y axis. The system specific details illustrated in Figure 8 are shown in Table 3.

5 Results and discussion

The simulation results were achieved using sequential quadratic programming (SQP) available in MATLAB® toolbox. The SQP algorithm uses the penalty method proposed by [83, 84], where the penalty parameter for k^{th} iteration r_{k_i} corresponding to i^{th} constraint



Fig. 9 Cylindrical joint in Figure 6: Residuals Equation of motion (Color figure online).



Fig. 10 Cylindrical joint in Figure 6: X and Y coordinates of the centers of left and right ends (Color figure online).

is given as follows:

$$r_{k_i} = \max(\lambda_i, \frac{\lambda_i + r_{k-1_i}}{2}), \tag{43}$$

where λ_i is the Lagrange multiplier associated with i^{th} constraint.

The case-studies of cylindrical and prismatic joints were conducted with time step h varying between $h_{\min} = 10^{-3}$ and $h_{\max} = 10^{-2}$ seconds, while the revolute joint case study was conducted with h varying between $h_{\min} = 10^{-3}$ and $h_{\max} = 2 \times 10^{-3}$ seconds. Further, to study the convergence and stability properties and to highlight the effects of numerical damping, the total mechanical energy in the pendulum (revolute joint with clearance) was studied with different ranges of step sizes. The SQP algorithm comes with the advantage of fast convergence and its ability to efficiently deal with a large number of constraints [85]. The constraint violation was observed $\|\mathbf{\Phi}^{\text{ineq}}(\mathbf{q})\| \leq 10^{-10}$ upon the contact formation. At the same time the equations of motion (16a) and (16b) were satisfied within the limits of 10^{-9} .

5.1 Cylindrical joint

As shown in Figure 9, the equations of motion residuals were found to be of order 10^{-9} . Figure 10 shows the positions of X and Y components of the left-hand and the right-hand ends of the slider. As can be expected, the external force *F* generates torques and tilts the slider. Hence, the left end descends down and makes contact with the shaft, while the right end moves up to make contact with the shaft. After the contact is formed at the diametrically



Fig. 11 Cylindrical joint in Figure 6: Contact forces against time (Color figure online).



Fig. 12 Cylindrical joint in Figure 6: X and Y velocities of center of mass (Color figure online).



Fig. 13 Cylindrical joint in Figure 6: Z velocity of center of mass.

opposite ends of the rod j, the slider starts moving in the Z direction. Figure 11 shows the contact forces CF - x, CF - y, and CF - z in X, Y, and Z directions respectively. Figure 12 represents the X and Y velocities of the center of mass. The lateral velocities (in X-Y directions) gained by slider are very small, though periodically varying. These oscillations of low magnitude arise because in the absence of friction, minute perturbations in X-Y plane can rotate the slider about Z axis, which also happens to be the major axis of the slider. Figure 13 represents the Z-velocity of the center of mass of the slider, and Figure 14 represents the displacement of slider Z direction from the initial position. Since the external force is constant, the acceleration is constant, and hence the velocity and position profiles are straight-line and parabola respectively.

Figure 15 shows the step size adaptation over time. It can be observed that the dramatic fluctuations in the step size correlate well with the lateral micro-oscillations of the contact



Fig. 14 Cylindrical joint in Figure 6: Z coordinate of center of mass.



Fig. 15 Cylindrical joint in Figure 6: Time step size.



Fig. 16 Prismatic joint in Figure 7: Equations of motion residuals.

points that undergo sliding while maintaining the contact, as observed in Figure 10. In cylindrical slider, the rotational degree of freedom about Z axis is free and remains so even after contact is formed. Moreover, the torque generated by external force is acting about the intermediate axis of the slider. The torque generated by external force is acting throughout the simulation. Since rotation about intermediate axis is unstable to numerical perturbations along global X and Y [78], the step size also oscillates between the bounds in this case.



Fig. 17 Prismatic joint in Figure 7: Y coordinates of contact points P_{f1} and P_{b1} (Color figure online).



Fig. 18 Prismatic joint in Figure 7: Contact forces against time (Color figure online).

5.2 Prismatic joint

The equations of motion residuals, as shown in Figure 16, are in order of 10^{-9} . Figure 17 shows the positions of Y components of the points P_{f_1} and P_{b_1} shown in Figure 7. The external force F generates a torque and tilts the slider, similar to the cylindrical joint's case. However, the contact happens on all the six vertices. After the contact is formed at the body *j*, the slider starts moving in the Z direction. Figure 18 shows the contact forces CF - x, CF - y and CF - z in X, Y, and Z directions respectively. Figure 19 represents the X and Y velocities of the center of mass. Once the steady contact is formed (around t = 0.2s), the center of mass does not move in X-Y plane. Figure 20 represents the Z-velocity of the center of mass, and Figure 21 represents the motion of slider's center of mass along Z direction. Since the external force is constant, the acceleration is constant, and hence the velocity profile is straight line and the position profile is parabolic. The Figure 22 represents the step size adaptations against time. In case of prismatic slider, the external torque (by the applied force) is acting about the intermediate axis, which eventually tilts the slider until the contact is made at the vertices such that the only remaining degree of freedom remains sliding along the Z axis. It can be observed that once a steady contact is maintained, the step size also stabilizes to the maximum value of 10^{-2} . The main reason why Figure 15 is noisy and Figure 22 is not is that in prismatic slider, rotational perturbations about Z cannot destabilize the solution as formation of contact at vertices immediately nullifies that degree of freedom.

5.3 Revolute joint

In this section, the results obtained from the proposed methodology have been discussed for the revolute joint case study. The counterparts from an ideal joint case, i.e., without any



Fig. 19 Prismatic joint in Figure 7: X and Y velocities of center of mass (Color figure online).



Fig. 20 Prismatic joint in Figure 7: Z velocity of center of mass.



Fig. 21 Prismatic joint in Figure 7: Z position of center of mass.

clearance, have also been compared. The ideal joint case was evaluated using tangent space ODE (TSODE) formulation [75]. For brevity, the term "NLP" represents the nonsmooth solution of the respective system with clearance. The rationale behind choosing a separate well-established methodology for simulating an ideal joint case was to check the credibility of the solution obtained from the method proposed in this work.

Figure 23 shows the residuals of equations of motion. Figure 24 shows the positions of X and Y components of the contact points on the hinge pin's surface. It can be observed that because of rotation the system undergoes intermittent formation and breaking off of the contact. Figure 25 shows the contact forces CF - x, CF - y, and CF - z in X, Y, and Z directions respectively. The counterparts from an ideal joint case, i.e., without any clearance, have also been compared. It can be observed that the general pattern of the variation of the contact forces is sinusoidal, excluding the peaks. When juxtaposed to the ideal-joint case, a



Fig. 22 Prismatic joint in Figure 7: Time step size.



Fig. 23 Revolute joint in Figure 8: 2-norm residuals of equations of motion



Fig. 24 Revolute joint in Figure 8: X and Y coordinates of the centers of left and right ends (Color figure online).

similar pattern can be observed; however, with a slight lag. This lag can be attributed to the fact that the pendulum, when released, first falls as a free body and consumes the clearance. Then the contact is made and the expected motion begins. Figure 26 represents the X and Y velocities of the center of mass.

The effect of lag can also be observed in the initial moments of time in Figure 26, while the rest of the velocity profiles overlap the ideal-joint counterparts. This overlap can also be observed in the position profiles. Figure 27 represents the positions of center of mass



Fig. 25 Revolute joint in Figure 8: Contact forces against time. Joint with clearance formulated as NLP (25). Ideal joint case formulated as a TSODE [75] (Color figure online).



Fig. 26 Revolute joint in Figure 8: X and Y velocities of center of mass (Color figure online).



Fig. 27 Revolute joint in Figure 8: X and Y coordinates of center of mass (Color figure online).

in X-Y plane. In Figure 28, the axial displacement of the center of mass of the pendulum is shown evolving with time. As can be expected, the ideal-joint case does not exhibit any displacement along the Z direction. Furthermore, it must be noticed that the center of mass is located at a significant distance from the axis of rotation. Hence, the displacement of center of mass along Z is a combination of the tilting effect on the hinge pin because of radial clearance as well as the prismatic displacement in the Z direction because of axial clearance.

Figure 29 represents the variation of step size with time. It can be observed that step size changes when the contact points acquire high relative sliding speeds. Moreover, the nu-



Fig. 28 Revolute joint in Figure 8: Z coordinates of center of mass (Color figure online).



Fig. 29 Revolute joint in Figure 8: Time step size variation in NLP solution of the system with clearance.

| specifications: Revolute joint | Property | Value | Units | |
|--------------------------------|----------|-------|-------|--|
| | m | 2 | kg | |
| | l | 20 | m | |
| | R | 5 | m | |
| | cm_X | 30.9 | m | |
| | С | 0.1 | m | |
| | | | | |

merical damping because of large range of step size happens to be a source of dissipation. This is confirmed in Figure 30 where the total mechanical energy is plotted for different ranges of step size. Recall that our base-line step size range was $10^{-3} < h < 2 \times 10^{-3}$. The convergence analysis shows that limiting the range to smaller values of maximum step size $(5 \times 10^{-4} < h < 10^{-3})$ resulted in significant improvement in energy conserving properties. Further, the stability analysis compares the two baseline convergence ranges with their respective increased and decreased range, keeping the maximum step size same, i.e., 10^{-3} and 2×10^{-3} respectively. It can be observed that energy loss is expected with larger step sizes, but it can be restored using smaller step sizes. Further, with the maximum step size fixed, improved stability comes with smaller minimum step size.

5.4 Ideal joint pendulum

Figure 31 shows a simple case study of a pendulum *i* forming a revolute joint with a hinge pin *j* of radius *R*. The center of mass of the pendulum is located at cm_x along the local X direction. The pendulum *i* has mass *m* with its center of mass denoted by the generalized



Fig. 30 Revolute joint in Figure 8: Stability and convergence properties for different ranges of step size (Color figure online).



Fig. 31 Schematic diagram of a pendulum demonstrating a revolute joint (Color figure online)

coordinates $\mathbf{q}_i \in \mathbb{R}^7$ and the hinge pin *j* is mass-less and fixed at both ends. The gravity acts along the $-\mathbf{Y}$ axis and the clearance in the revolute joint is *c*. The system specific details illustrated in Figure 31 are shown in Table 4. Two cases of the showed pendulum are considered: 1. Joint with clearance using NLP (25); 2. Ideal joint using NLP (31). Further, the fixed step size studies were conducted with two step size values: $h = 2 \times 10^{-4}$ and $h = 10^{-3}$ for both cases. The effect of step size range bounds was also compared by simulated



Fig. 32 Pendulum in Figure 31: The center of mass position represented in X-Y coordinates (Color figure online)



Fig. 33 Pendulum in Figure 31: Center of mass velocity represented as the rate of change of X-Y coordinates (Color figure online)



Fig. 34 Pendulum in Figure 31: Total, kinetic, and potential energies (Color figure online)

both cases with two ranges of bounded step size: $2 \times 10^{-4} \le h \le 10^{-3}$ and $10^{-3} \le h \le 2 \times 10^{-3}$. Based on inferences from Figure 30, the aim is to investigate the effect of numerical damping on entities other than mechanical energy.

Figures 32 and 33 respectively compare the X-Y positions and velocities of the center of mass. The effect of clearances is more noticeable in velocity plots, especially in the beginning. Figure 34 compares the total mechanical energy and its kinetic and potential



Fig. 35 Pendulum in Figure 31: Contact forces along global X, Y, and Z directions. Large step size induces numerical damping that reduces the number of peaks (Color figure online).



Fig. 36 Pendulum in Figure 31 with ideal joint: Norm of kinematic constraints (Color figure online).

components for the two cases, where only gravitational potential energy was considered to calculate total mechanical energy. The nonsmooth behavior in total energy can be observed in the interval 0 < t < 2.

Figure 35 compares the variation in spatial components of contact forces between clearance joint and ideal joint, both cases simulated by the discussed NLP approach. In Figure 35, it can be observed that numerical damping comes into effect with large step sizes as the number of peaks go down with range of larger step sizes. Figure 36 shows that the formulation does satisfy the kinematic constraints in case of ideal joint. The plot in the top represents the X and Y coordinates of left and right ends of the pivot end's centerline. It can be noticed that the clearance constraint of 0.1 m is not violated. In case of ideal joint, the constraints are satisfied well. This case study demonstrates the effectiveness of the proposed framework to deal with smooth as well as nonsmooth systems. Further, Figure 37 compares the com-



Fig. 37 Pendulum in Figure 31: Total iterations between t = 0s and t = 10s, and computational time (Color figure online).

putational aspects of the two cases studied in this section. The advantage of bounded step size over fixed step size can hence be argued based on the observations in case of joints with clearances. However, there is no significant improvement in case of ideal joints if the h_{min} is kept same, i.e., 10^{-4} or 10^{-3} . However, increasing the step size range indeed reduces the computation time and induces numerical damping, at the cost of energy loss, as observed in Figure 30. Moreover, it must be noticed that contact problems consume significantly more computational resources than smooth problems.

6 Conclusions

This work provides a nonsmooth dynamics based framework for the purpose of simulating multibody systems with clearances in the joints. Nonsmooth dynamics of spatial cylindrical, prismatic, and revolute joints with clearances have been derived assuming inelastic nature of the contact. The mathematical representations of constraint inequalities have been carefully derived to ensure that they capture all the possible scenarios of contact formation. Simulation results demonstrate the computational effectiveness of the proposed framework. Moreover, the tolerance limits of constraints during contact are also satisfied within significant accuracy. The proposed framework was also successfully applied to simulate ideal joints in **rA** formulation, and constraint requirements were also well preserved.

The paper proposes a novel adaptive time step strategy based on treating the time step size h as a variable in the NLP problem. To preserve rotations, normalization constraint on Euler parameters was enforced. As a result, the velocity of generalized coordinates $\dot{\mathbf{q}}$ was included as an optimization variable and the integration scheme was incorporated within the optimization process. The stability analysis showed that the deviation in total energy can be mitigated by selecting appropriate bounds on the step size. Smaller values of minimum bounds were found conducive to reduce energy deviation. Furthermore, looking at the level of complexity of problems considered and the computational time consumed, a limitation of the proposed framework can be observed. Indeed, this is an open task for future work.

This work focuses on frictionless contacts in the joints with clearances. On-going work by the authors seeks to extend this study to the simulation of joints with clearance and friction, which will shed further light on the impact of clearances on multibody system dynamics.

The industrial approach of assigning tolerances, and hence clearances, is primarily based on manufacturability related aspects. However, the effects of accumulated clearances on the kinematic and dynamic parameters of a mechanical system are often neglected. The presented work can be impactful in rationalizing the design procedures because tolerance assignment is a critical task on engineering drawings. Understanding the effect of tolerances on the system's dynamic response can be helpful in optimizing the design of a machine's components.

Acknowledgements This work was supported by the Terramechanics, Multibody, and Vehicle Systems (TMVS) Laboratory and Computational Science Laboratory (CSL), Virginia Tech, Blacksburg, VA, USA. The authors express their deep gratitude to the anonymous reviewers and editor whose suggestions assisted greatly in improving the quality of paper.

Author contributions E. C. did the conceptualization, methodology, code implementation, results analysis and documentation. C. S. provided supervision and technical expertise in multibody dynamics. A. S. provided supervision and helped develop the proposed methodology. All the authors equally contributed to writing, responding to the reviewers' comments, and revising-editing of the manuscript.

Declarations

Competing interests The authors declare no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Hertz, H.R.: Ueber die Beruehrung fester elastischer Koerper. J. fuer die reine und angewandte Mathematik 91, 156–171 (1882). For English translation see Miscellaneous Papers by H. Hertz, edited by D.E. Jones and G.E. Schott (Macmillan, London, 1896)
- 2. Goldsmith, W.: Impact: The Theory and Physical Behaviour of Colliding Solids. Arnold, Sevenoaks (1960)
- Hunt, K.H., Crossley, F.R.E.: Coefficient of restitution interpreted as damping in vibroimpact. J. Appl. Mech. 7, 440–445 (1975)
- Lankarani, H.M., Nikravesh, P.E.: Continuous contact force models for impact analysis in multibody systems. Nonlinear Dyn. 5(2), 193–207 (1994)
- Lankarani, H.M., Nikravesh, P.E.: A contact force model with hysteresis damping for impact analysis of multibody systems. J. Mech. Des. 112(3), 369–376 (1990)
- Radzimovsky, E.I.: Stress Distribution and Strength Condition of Two Rolling Cylinders Pressed Together, vol. 408. University of Illinois (1953)
- Dubowsky, S., Freudenstein, F.: Dynamic analysis of mechanical systems with clearances-Part 1: Formation of dynamic model. J. Eng. Ind. 93(1), 305–309 (1971)
- 8. Johnson, K.L.: Contact Mechanics. Cambridge University Press, London (1995)
- Pereira, C., Ramalho, A., Ambrosio, J.: An enhanced cylindrical contact force model. Multibody Syst. Dyn. 35(3), 277–298 (2015)
- Skrinjar, L., Slavic, J., Boltezar, M.: A review of continuous contact-force models in multibody dynamics. Int. J. Mech. Sci. 145, 171–187 (2018)
- Tian, Q., Flores, P., Lankarani, H.M.: A comprehensive survey of the analytical, numerical and experimental methodologies for dynamics of multibody mechanical systems with clearance or imperfect joints. Mech. Mach. Theory 122, 1–57 (2018)
- Flores, P., Lankarani, H.M.: Contact Force Models for Multibody Dynamics, 1st edn. Springer, Cham (2018)
- Corral, E., Moreno, R., Meneses, J., Garcia, M., Castejón, C.: Spatial algorithms for geometric contact detection in multibody system dynamics. Mathematics 9, 1359 (2021). https://doi.org/10.3390/ math9121359

- Dopico, D., Luaces, A., Saura, M., Cuadrado, J., Vilela, D.: Simulating the anchor lifting maneuver of ships using contact detection techniques and continuous contact force models. Multibody Syst. Dyn. 46, 147–179 (2019)
- Dopico, D., Gonzalez, F., Cuadrado, J., Kövecses, J.: Determination of holonomic and non-holonomic constraint reactions in an index-3 augmented Lagrangian formulation with velocity and acceleration projections. J. Comput. Nonlinear Dyn. 9, 041006 (2014)
- Choi, J., Ryu, H., Kim, C., Choi, J.: An efficient and robust contact algorithm for a compliant contact force model between bodies of complex geometry. Multibody Syst. Dyn. 23, 99 (2010)
- Nikravesh, P., Lankarani, H.: A contact force model with hysteresis damping for impact analysis of multibody systems. J. Mech. Des. 112, 369–376 (1990)
- Shen, Y., Xiang, D., Wang, X., Wei, Y.: A contact force model considering constant external forces for impact analysis in multibody dynamics. Multibody Syst. Dyn. 44, 397–419 (2018)
- Sharf, I., Zhang, Y.: A contact force solution for non-colliding contact dynamics simulation. Multibody Syst. Dyn. 16, 263–290 (2006)
- Zakhariev, E.: Dynamics of rigid multibody systems with clearances in the joints. Mech. Struct. Mach. 27, 63–87 (1999)
- Ibrahimi, S., Salahshoor, E., Nouri, S.: Sensitivity analysis for optimal design of multibody systems with clearance joint. Int. J. Manuf. Technol. Manag. 11, 35–44 (2018)
- Flores, P., Lankarani, H.: Dynamic response of multibody systems with multiple clearance joints. J. Comput. Nonlinear Dyn. 7, 031003 (2012)
- Xiang, W., Yan, S., Wu, J., Niu, W.: Dynamic response and sensitivity analysis for mechanical systems with clearance joints and parameter uncertainties using Chebyshev polynomials method. Mech. Syst. Signal Process. 138, 106596 (2020)
- Bauchau, O., Rodriguez, J., Bottasso, C.: Modeling of unilateral contact conditions with application to aerospace systems involving backlash, freeplay and friction. Mech. Res. Commun. 28, 571–599 (2001)
- Moreau, J.J.: Unilateral contact and dry friction in finite freedom dynamics. In: Moreau, J.J., Panagiotopoulos, P.D. (eds.) Non-smooth Mechanics and Applications. CISM Courses and Lectures, vol. 302, pp. 1–82. Springer, Wien (1988)
- Moreau, J.J.: Application of convex analysis to some problems of dry friction. In: Zorski, H. (ed.) Trends in Applications of Pure Mathematics to Mechanics, vol. 2. Pitman, London (1979)
- Moreau, J.J.: Une Formulation du Contact a Frottement Sec; Application Au Calcul Numerique. C.R. Acad. Sci., Paris (1986). Technical Report 13, Serie II
- 28. Panagiotopoulos, P.D.: Inequality Problems in Mechanics and Applications. Birkhäuser, Boston (1985)
- 29. Panagiotopoulos, P.D.: Hemivariational Inequalities. Springer, Berlin (1993)
- Monteiro Marques, M.D.P.: Differential Inclusions in Nonsmooth Mechanical Problems. Birkhäuser, Basel (1993)
- Acary, V., Brogliato, B.: Numerical Methods for Nonsmooth Dynamical Systems. Lecture Notes in Applied and Numerical Mechanics. Springer, Berlin (2008)
- 32. Brogliato, B.: Nonsmooth Mechanics. Springer, Heidelberg (1999)
- Leine, R., Nijmeijer, H.: Dynamics and Bifurcations of Non-smooth Mechanical Systems. Springer, Berlin (2004)
- Jean, M., Moreau, J.J.: Unilaterality and dry friction in the dynamics of rigid body collections. In: Curnier, A. (ed.) Proceedings of Contact Mechanics International Symposium, pp. 31–48. EPFL, Lausanne (1992)
- 35. Brogliato, B.: Nonsmooth Impact Dynamics. Springer, London (1996)
- 36. Fremond, M.: Non-smooth Thermomechanics. Springer, Berlin (2002)
- 37. Glocker, C.: Set-Valued Force Laws Dynamics of Nonsmooth Systems. Springer, Berlin (2001)
- 38. Pfeiffer, F.: Mechanical System Dynamics. Springer, Berlin (2008)
- 39. Wriggers, P.: Computational Contact Mechanics. Springer, Berlin (2006)
- 40. Murty, K.G.: Linear complementarity, linear and nonlinear programming. In: White, D.J. (ed.) Sigma Series in Applied Mathematics. Heldermann Verlag, Berlin (1988)
- Klarbring, A.: Mathematical programming and augmented Lagrangian methods for frictional contact problems. In: Curnier, A. (ed.) Proceedings of Contact Mechanics International Symposium, pp. 409–422. EPFL, Lausanne (1992)
- 42. Klarbring, A.: Mathematical programming in contact problems. In: Aliabadi, M.H. (ed.) Computational Methods for Contact Problems. Elsevier, Amsterdam (1994)
- Mangasarian, O.L.: Equivalence of the complementarity problem to a system of nonlinear equations. SIAM J. Appl. Math. 31(1), 89–92 (1976)
- Cosimo, A., Galvez, J., Cavalieri, F.J., Cardona, A., Brüls, O.: Equivalence of the complementarity problem to a system of nonlinear equations. SIAM J. Appl. Math. 31(1), 89–92 (1976)

- Cosimo, A., Galvez, J., Cavalieri, F.J., Cardona, A., Brüls, O.: On the adaptation of local impact laws for multiple impact problems. Nonlinear Dyn. 102, 1997–2016 (2020)
- Brogliato, B., ten Dam, A.A., Paoli, L., et al.: Numerical simulation of finite dimensional multibody nonsmooth mechanical systems. Appl. Mech. Rev. 55(2), 107–150 (2002)
- 47. Bremer, H.: Dynamik und Regelung Mechanischer Systeme. Teubner Studienbücher, Mechanik, B.G. Teubner, Stuttgart (1988)
- 48. Bremer, H., Pfeiffer, F.: Elastische Mehrkörpersysteme. Teubner, Stuttgart (1992)
- 49. Pfeiffer, F.: Mechanische Systeme mit unstetigen Übergängen. Ing.-Arch. 54, 232–240 (1984)
- Pfeiffer, F.: Multibody dynamics with multiple unilateral contacts. In: Proceedings of the XIXth International Congress of Theoretical and Applied Mechanics, Kyoto, Japan (1996)
- Pfeiffer, F., Glocker, Ch.: Multibody Dynamics with Unilateral Contacts. Wiley Series in Nonlinear Science. Wiley, New York (1996)
- Ch, G.: Dynamik von Starrkrpersystemen mit Reibung und Stößen. Fortschritt-Berichte VDI 18: Mechanik/Bruchmechanik (182). VDIVerlag, Düsseldorf (1995)
- Glocker, Ch., Pfeiffer, F.: Multiple impacts with friction in multibody systems. Nonlinear Dyn. 7, 471–497 (1995)
- Beitelschmidt, M.: Reibstöße in Mehrkörpersystemen. Fortschritt-Berichte VDI 11(275). VDI-Verlag, Düsseldorf (1999)
- Jean, M.: The non-smooth contact dynamics method. Comput. Methods Appl. Mech. Eng. 177, 235–257 (1999)
- Silcowitz-Hansen, M., Niebe, S., Erleben, K.: A nonsmooth nonlinear conjugate gradient method for interactive contact force problems. Vis. Comput. 26, 893–901 (2010)
- Anitescu, M.: Optimization-based simulation of nonsmooth rigid multibody dynamics. Math. Program. 105(1), 113–143 (2006)
- Negrut, D., Serban, R., Tasora, A.: Posing multibody dynamics with friction and contact as a differential complementarity problem. J. Comput. Nonlinear Dyn. 13, 014503 (2018)
- Orden, J.C.G.: Analysis of joint clearances in multibody systems. Multibody Syst. Dyn. 13, 401–420 (2005)
- Ambrósio, J., Pombo, J.: A unified formulation for mechanical joints with and without clearances/bushings and/or stops in the framework of multibody systems'. Multibody Syst. Dyn. 42(3), 317–345 (2018)
- Flores, P., Ambrósio, J., Claro, J.P.: Dynamic analysis for planar multibody mechanical systems with lubricated joints. Multibody Syst. Dyn. 12, 47–74 (2004)
- Liu, Q., Lu, S.N., Ding, X.L.: An error equivalent model of revolute joints with clearances for antenna pointing mechanisms. Chin. J. Mech. Eng. 31, 39 (2018)
- 63. Marques, F., Isaac, F., Dourado, N., Flores, P.: 3D formulation for revolute clearance joints. In: New Trends in Mechanism and Machine Science, Nantes, France, pp. 20–23 (2016)
- Marques, F., Isaac, F., Dourado, N., Flores, P.: An enhanced formulation to model spatial revolute joints with radial and axial clearances. Mech. Mach. Theory 116, 123–144 (2017)
- Flores, P., Ambrósio, J., Claro, J.C.P., Lankarani, H.M., Koshy, C.S.: A study on dynamics of mechanical systems including joints with clearance and lubrication. Mech. Mach. Theory 41(3), 247–261 (2006)
- Flores, P., Ambrósio, J., Claro, J.C.P., Lankarani, H.M.: Dynamics of multibody systems with spherical clearance joints. J. Comput. Nonlinear Dyn. 1(3), 240–247 (2006)
- Flores, P., Leine, R., Glocker, C.: Modeling and analysis of planar rigid multibody systems with translational clearance joints based on the non-smooth dynamics approach. Multibody Syst. Dyn. 23, 165–190 (2010)
- Cavalieri, F.J., Cardona, A.: Non-smooth model of a frictionless and dry three-dimensional revolute joint with clearance for multibody system dynamics. Mech. Mach. Theory 121, 335–354 (2018)
- Akhadkar, N., Acary, V., Brogliato, B.: Multibody systems with 3d revolute joints with clearances: an industrial case study with an experimental validation. Multibody Syst. Dyn. (2017)
- Akhadkar, N., Acary, V., Brogliato, B.: 3D revolute joint with clearance in multibody systems. In: Comput. Kinem., Mech. Mach. Sci., vol. 50 (2018)
- Flores, P., Ambrósio, J., Claro, J.C.P., Lankarani, H.M.: Translational joints with clearance in rigid multibody systems. J. Comput. Nonlinear Dyn. 3(1), 011007 (2008)
- Anitescu, M., Tasora, A.: An iterative approach for cone complementarity problems for nonsmooth dynamics. Comput. Optim. Appl. 47, 207–235 (2010)
- Anitescu, M., Hart, G.D.: A constraint-stabilized time-stepping approach for rigid multibody dynamics with joints, contact and friction. Int. J. Numer. Methods Eng. 60(14), 2335–2371 (2004)
- Todorov, E.: Convex and analytically-invertible dynamics with contacts and constraints: theory and implementation in mujoco. In: 2014 IEEE International Conference on Robotics and Automation (ICRA), pp. 6054–6061 (2014). https://doi.org/10.1109/ICRA.2014.6907751

- Haug, E.J.: Computer-aided kinematics and dynamics of mechanical systems: Volume II: Modern Methods, ResearchGate (2022). www.researchgate.net
- Kissel, A., Negrut, D., Taves, J.: On an exponential map approach for rigid body kinematics and dynamics analysis (2020). Technical Report TR-2020-08. https://sbel.wisc.edu/technicalreports/
- 77. Kissel, A., Negrut, D., Taves, J.: Dwelling on the connection between SO(3) and rotation matrices in rigid multibody dynamics. Part 1: Description of an index-3 DAE solution approach Proceedings of the ASME 2021 International Design Engineering Technical Conferences and Computers and Information in Engineering Conference IDETC-CIE2021
- Meriam, J.L., Kraige, L.G., Bolton, J.N.: Engineering Mechanics: Dynamics, 8th edn. Wiley, New York (2015)
- Udwadia, F.E.: The general Gauss principle of least constraint. J. Appl. Mech. 90(11), 111006 (2023). https://doi.org/10.1115/1.4062887
- Yang, L., Xue, S., Yao, W.: Application of Gauss principle of least constraint in multibody systems with redundant constraints. Proc. Inst. Mech. Eng., Part K, J. Multi-Body Dyn. 235(1), 150–163 (2021)
- Frâncu, M., Moldoveanu, F.: Unified simulation of rigid and flexible bodies using position based dynamics. In: Eurographics Proceedings: Workshop on Virtual Reality Interaction and Physical Simulation VRIPHYS (2017)
- Anitescu, M., Tasora, A.: A matrix-free cone complementarity approach for solving large-scale, nonsmooth, rigid body dynamics. Comput. Methods Appl. Mech. Eng. 200, 439–453 (2011)
- Han, S.P.: A globally convergent method for nonlinear programming. J. Optim. Theory Appl. 22, 297 (1977)
- Powell, M.J.D.: A Fast Algorithm for Nonlinearly Constrained Optimization Calculations. In: Watson, G.A. (ed.): Numerical Analysis. Lecture Notes in Mathematics, vol. 630. Springer, Berlin (1978)
- Xiao, Z.: A Comparative Analysis of an Interior-point Method and a Sequential Quadratic Programming Method for the Markowitz Portfolio Management Problem. Undergraduate thesis, Oberlin College (2016)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.